Author's response to reviews

Title: Factors associated with having less than 20 natural teeth in rural adults: A cross-sectional study

Authors:

Su-Jen Tsai (sujent@ms38.hinet.net)
Wen-Nan Chou (wendy0625@cgmh.org.tw)
Liang-Tse Tu (final1024@adm.cgmh.org.tw)
Ming-Shyan Lin (ningshyan@gmail.com)
Mei-Yen Chen (meiwen@gw.cgust.edu.tw)

Version: 4 Date: 30 November 2015

Author's response to reviews:

Dear editor,

since the first author (Dr. Tsai) missed your letter and as a corresponding author I did not receive the latest information. Therefore, I upload this minor revision till today.

1. We have followed your comment that all tables are included at the end of the main manuscript.
2. In title page, we have provided some new information for the authorship with red font.